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  BEAM: 
 Beam intensity stabilized after Tue Fermi power outage. 

  NearDet:  
 It is up and running as cool, with full HV. 
 At beginning of last weak the timing issue, delayed in data. 
 Occurred after Tue power outage, Wed waiting for beam to 

confirm it’s solved.  
  FarDet:  
 It is up and running.  
 Physics (cosmics) 13-diblocks running. 
 Retrofitting is spread over through only 1 diblocks. 
 Big maintenance day on Tue – tightening of Dry Gas system 

  Control Room:  
 Used old (12th floor) CR for generator backup power 

maintenance on Fri and ROCW through weekend on backup 
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Operations I – summary  



FarDet: Data-Taking 
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Partition 1 (status last week) 
 diblocks 2 – 14  
 Running ~99% active,  
 ALL COOLED, HV at full gain. 
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Current status - Retrofitting: 
 Retrofit: ALL OLD APDs & brass fittings away! 
 Already retrofitted DiB-{04,05,03,02}. 
 New APDs installed on missing DiB-01-side 
 First dcm on side today, ongoing work 
affected by APDs available  
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FD: DAQ Uptime 

 Last weak running with 92.05% DAQ uptime, very good. 
 On Tue big maintenance day during beam shutdown (only 65% uptime):  

 tightening of one Dry Gas system, found and replaced 14 bad APDs 
 Replaced another 14 APDs showed very noisy, 100% efficient work. 
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FD: POTs delivered/recorded 

 Past weak efficiency at 96.9% - very good but low number of POTs delivered 
 Tue beam shutdown, bean issues on Wed, weekend good not best. 
 We recorded in average 4.7e18  (from 4.9) pots for past weak 
 Very good timing for maintenance on Tue (96% recorded) 
 Bad Wed (51%), however absolute POTs very low. 


	NOvA Experiment Status�All Experimenter’s Meeting
	Operations I – summary 
	FarDet: Data-Taking
	FD: DAQ Uptime
	FD: POTs delivered/recorded

